Author's response to reviews

Title: The NKF-NUS hemodialysis trial protocol - a randomized controlled trial to determine the effectiveness of a self management intervention for hemodialysis patients

Authors:

Konstadina Griva (psygk@nus.edu.sg)
Nandakumar Moopil (nandakumar.m@nkfs.org)
Penny Seet (penny.seet@nkfs.org)
Deby Sarojuv Pala Krishnan (ds.palakrishnan@nkfs.org)
Hayley James (hayley.james.1@city.ac.uk)
Stanton P Newman (Stanton.Newman.1@city.ac.uk)

Version: 4 Date: 25 January 2011

Author's response to reviews:

Correction of minor typographic errors